CS 261, Fall 2004 Computer Security

Instructor: David Wagner daw@cs Lectures:

Office Hours:

Course Description CS261: Security in Computer Systems

systems and networking. Course topics Grading Class project: 45% Problem sets: 40% Paper reviews and class discussion: 15%

Projects

Problem Sets

semester, to appear on the course webpage as they are assigned.

accepted.

Readings

each

Warning not expected to behave responsibly.

security hole is no excuse. Feedback

this one

daw@cs.berkeley.edu http://www.cs.berkeley.edu/~daw/